Author's response to reviews

Title: Coronary artery bypass grafting and sensorineural hearing loss, a cohort study.

Authors:

Omer Ashraf (warraicch@yahoo.com)

Version: 5 Date: 4 December 2005

Author's response to reviews:

Sir,

Following changes have been made in the manuscript titled 'Coronary artery bypass grafting and sensorineural hearing loss, a cohort study', as part of final formatting requirements. Abstract has now been divided into Background, Methods/Design and Discussion sections. Manuscript has also been divided into Abstract; Background; Methods/Design; Discussion; Abbreviations; Competing interests; Authors' contributions and References, sections in the appropriate order.

Regards.